Author’s response to reviews

Title: Deflazacort: therapeutic index, relative potency and equivalent doses versus other corticosteroids

Authors:
Luca Parente (lparente@unisa.it)

Version: 2 Date: 25 Oct 2016

Author’s response to reviews:

Fisciano, October 25th 2016

Editorial Office
BMC Pharmacology and Toxicology

Dear Sirs

Subject: Resubmission of manuscript # PHAT-D-16-00102R1

Deflazacort: therapeutic index, relative potency and equivalent doses versus other corticosteroids

Many thanks for your mail letter of October 24th.

I am writing to resubmit the manuscript. Please find herewith enclosed a report on all the changes made according to the comments indicated in the mail letter.

1. In the declarations section it is now stated that no funding was available. See page 26, line 2
2. A list of figures has been inserted after the references. See page 28, lines 38-39
Hoping that the manuscript is acceptable for publication in its present form I look forward to hearing from you.

Sincerely

LUCA PARENTE

Professor of Pharmacology and Pharmacotherapy

Address for correspondence

Luca Parente

Department of Pharmacy, University of Salerno

Via Giovanni Paolo II, 132 - 80084 Fisciano (SA) - Italy

Phone: +39 089 969760

e-mail: lparente@unisa.it